Author's response to reviews

Title: Development of an integrated system for classification, assessment and comparison of services for Long-Term Care in Europe: The eDESDE-LTC Study

Authors:

Luis Salvador-Carulla (luis.salvador@telefonica.net)
Cristina Romero (cristina.romero@uca.es)
Mencia R Gutierrez-Colosia (mencia.ruiz@gmail.com)
Sonia Johnson (s.johnson@ucl.ac.uk)
Germain Weber (germain.weber@univie.ac.at)
David McDaid (D.Mcdaid@lse.ac.uk)
Hristo Dimitrov (dvchristo2001@yahoo.com)
Lilijana Sprah (lilijana.sprah@guest.arnes.si)
Birgitte Kalseth (birgitte.kalseth@sintef.no)
Giuseppe Tibaldi (btibaldi@hotmail.com)
Jose A. Salinas-Perez (jsalinas@uloyola.es)
Carolina Lagares-Franco (carolina.lagares@uca.es)
Maria Teresa Roma-Ferrí (mtr.ferri@ua.es)
Javier Alvarez-Galvez (javieralvarezgalvez@gmail.com)

Version: 2 Date: 16 July 2012

Author's response to reviews: see over
Recommendations

1. **Transference of the paper to the journal BMC Health Service Research**
   The authors agree the paper be transferred to the journal BMC Health Service Research

Modifications

2. **Competing interest**
   The ‘competing interest’ section has been included in the paper.